Author's response to reviews

Title: A gene expression profile for detection of sufficient tumour cells in breast tumour tissue: microarray diagnosis eligibility

Authors:

Paul Roepman (paul.roepman@agendia.com)
Arenda Schuurman (arenda.schuurman@agendia.com)
Leonie JMJ Delahaye (leonie.delahaye@agendia.com)
Anke T Witteveen (anke.witteveen@agendia.com)
Arno N Floore (arno.floore@agendia.com)
Annuska M Glas (annuska.glas@agendia.com)

Version: 4 Date: 5 August 2009

Author's response to reviews:

Dear Editorial Production team,

As per your request I placed the legend text if the tables below the table instead of above the Table.

I also removed a typographical error in the section Authors contributions "and" in stead of "ans".

Furthermore throughout the manuscript I consequently changed histo-pathological to "histopathological" (but versions of the word were in the manuscript).

All other requests were ignored since they were part of another manuscript (email 5 aug 2009, Philip Dooner).

I hope this solves all issues, please let me know if there is anything that I forgot.

Kind regards

Annuska Glas